

Received June 26, 2021, accepted July 24, 2021, date of publication August 3, 2021, date of current version August 18, 2021. *Digital Object Identifier* 10.1109/ACCESS.2021.3102033

Adaptive Queue Management and Traffic Class Priority Based Fairness Rate Control in Wireless Sensor Networks

SUBHA KANTA SWAIN[®], (Member, IEEE), AND PRADIPTA KUMAR NANDA, (Senior Member, IEEE)

Image and Video Analysis Laboratory, Department of Electronics and Communication Engineering, ITER, Siksha 'O' Anusandhan (Deemed to be University), Bhubaneswar 751030, India

Corresponding author: Subha Kanta Swain (subhakantaswain37@gmail.com)

ABSTRACT Congestion control in Wireless Sensor Networks (WSNs) is one of the key areas of research and different algorithms have been proposed using either of the notions of fair rate allocation, traffic class priority, and queue management. Use of the any one of the above is not adequate to address the challenges. Hence, in this paper, we have proposed a novel congestion control algorithm using the combined notions of fair allocation of bandwidth, prioritizing traffic classes, and Adaptive Queue Management (ADQM). The proposed Weighted Priority based Fair Queue Gradient Rate Control (WPFQGRC) scheme achieves the fair distribution of spare bandwidth by considering the traffic class priority, average queue size, and the connected loads of a node. Average queue size at every node is adapted based on the proposed notion of the gradient of the differential of Global Priority (GP) with respect to the differential of queue size. The output rate of a given node is computed based on the GP of the node and the average queue size. The spare bandwidth of a node is fairly distributed by taking into account the connected load of the given node. The proposed algorithm is developed to suit to a general topology of WSN, however for the sake of illustration, we have considered a tree topology network that deals with both Real-Time (RT) and Non-Real Time (NRT) traffic classes. The proposed algorithm is implemented in NS3 platform in Linux environment and the performance of the algorithm is evaluated in terms of throughput, packet loss, packet delay, traffic class patterns, node mobility, and the average queue size. The performance of the proposed algorithm is found to be superior to that of Yaghmaee et al.'s, Brahma et al.'s, Monowar et al.'s, Sarode et al.'s, Difference of Differentials Rate Control (DDRC), Weighted Priority based DDRC (WPDDRC), and Priority based Fairness Rate Control (PFRC) algorithms respectively.

INDEX TERMS Fair bandwidth distribution, adaptive queue management, rate control, traffic class priority, mobility, wireless sensor networks.

рт

Q

Dool Time

Queue size

NOMENCLATURE

NOMENCEATORE		N1	Keal Time
AQM	Active Queue Management	HNRT	High priority NRT
ADQM	Adaptive Queue Management	LNRT	Low priority NRT
ADQMU	ADQM Unit	MAC	Medium Access Control
CCU	Congestion Control Unit	NS3	Network Simulator 3
CDU	Congestion Detection Unit	MNRT	Medium priority NRT
FCU	Fairness Control Unit	NRT	Non-Real Time
GP	Global Priority	PAU	Priority Adjustment Unit
GP_w	Weighted GP	P_{TC}	Traffic Class Priority
		P_w	Weighted Priority
The associate editor coordinating the review of this manuscript and		P_{GE}	Geographical Priority
	_		

The associate editor coordinating the review of this manuscript and approving it for publication was Zihuai $\text{Lin}^{(D)}$.

QoS	Quality of Service
RAU	Rate Adjustment Unit
SP	Source Priority
W _{HNRT}	Weight of HNRT
W_{LNRT}	Weight of LNRT
W _{MNRT}	Weight of MNRT
WSNs	Wireless Sensor Networks

I. INTRODUCTION

Wireless Sensor Networks (WSNs) have multifold challenges [1] and with the advancement of technology WSN is able to effectively handle multimedia data such as image, video, and audio from the environment, and hence is known as Wireless Multimedia Sensor Networks (WMSNs) [2], [3]. Many sensor nodes with one or many base stations are able to make different topologies. WSNs have a wide range of applications such as environmental condition monitoring, patient health monitoring, structural health monitoring, animal habitat monitoring, defence surveillance, agricultural health monitoring, farm soil parameter testing, and industrial monitoring.

WSNs have many features [4], [5] including traffic characteristics, diverse applications, unique network topology, resource constraints, and small message size. The Quality of Service (QoS) is affected by limited power, lossy medium, limited resource, less storage capacity, dense deployment, low bandwidth utilization, and frequent change in topology, which results in low reliability, low energy efficiency and high congestion in the network [6]. Congestion is a crucial issue which occurs mainly due to collision, buffer overflow, and unfair transmission rate. This motivates many researchers to work on different congestion control and avoidance algorithms [7] to enhance the QoS of the network.

Multimedia data includes RT data which is gathered by the source nodes from the environment. Usually, the RT traffic is bursty [5], [6] in nature which requires high reliability and low latency, and hence prioritization is necessary for service differentiation. In this regard, the notion of traffic class priority is considered by Yaghmaee et al. [8] to propose a congestion control and rate control service differentiation model. In this model, they have considered four types of traffic classes such as RT traffic, low priority Non-Real Time traffic (NRT3), medium priority Non-Real Time traffic (NRT2), and high priority Non-Real Time traffic (NRT1) in separate queues. The performance of the algorithm is evaluated based upon low delay bound, high normalized throughput, and low loss probability. Yaghmaee et al. in their subsequent work [9] have used the notion of priority in monitoring the health conditions of different patients. Additionally, Swain et al. [10], [11] have also prioritized the traffic classes in their proposed rate control algorithms using their proposed notion of difference of differentials.

Besides the notion of traffic class priority, the notion of fairness has also been used to develop congestion control algorithms. Towards this end, an Interference-aware Fair Rate Control (IFRC) scheme is proposed by Rangwala *et al.* [12] by controlling the average queue size of topology with

multiple base stations. To tackle the issue of fair distribution of bandwidth, a rate-based Fairness Aware Congestion Control (FACC) protocol is proposed [13] based upon queue occupancy and hit frequency. Subsequently, a traffic management protocol, decoupling the congestion control module and fairness module is proposed by Brahma et al. [14] to improve the bandwidth utilization of the network. The authors have adaptively calculated the optimum transmission rate to allocate bandwidth to each traffic flow to achieve optimum throughput and fairness of each node in random topology. To increase the efficiency of the network based upon the notion of fairness, Kafi et al. [15] have proposed a Reliable, Efficient, Fair, and Interference-Aware Congestion Control (REFIACC) algorithm to control the wireless losses and interferences in a large sensor network. Recently, in our previous work [16], we have proposed a Priority-based Fairness Rate Control (PFRC) scheme where the fair allocation of bandwidth is achieved based upon traffic class priority, the connected load, and fixed queue size. In [16], authors have considered a fixed queue size for all the nodes at a given level of the network. The fixed queue size can not take care of the changing traffic pattern at different nodes and might onset congestion. Therefore, the queue size needs to be take care of different traffic patterns. Besides, the priority of the traffic class can be considered together with the fair allocation strategy to enhance the performance while avoiding congestion. This motivates us to pursue further research with the combined notion of the GP and fair allocation of bandwidth in the Adaptive Queue Management (ADQM) framework to enhance the throughput.

Though the combined notion of priority and fairness is able to address the issue of congestion to some extent, the congestion of a network is still influenced by the buffer overflow. To circumvent the problem of buffer overflow, Warrier et al. [17] have proposed a differential backlog algorithm using cross-layer optimization and the notion of queue differential. Further, queue-based congestion detection and rate control algorithm is proposed by Liang et al. [18] by evaluating the congestion level at each node. The issues of congestion and hotspot of body sensors are addressed by a rate control scheme proposed by Monowar et al. [19], where they have developed a temperature rise model jointly based on the queue occupancy and traffic intensity. It is found in the literature that performance of priority and fairness based rate control algorithms depends upon the queue size despite overcoming congestion to a certain degree. It is also found from the literature that the use of either the notions of traffic class priority or fairness control could overcome the issue of congestion to a certain extent.

In a network of mixed traffic, both prioritization of traffic class and the fair allocation of bandwidth should affect the queue size, which in turn will control congestion. Thus, queue size at every node, if adaptively managed, may lead to congestion avoidance due to varieties of the traffic class. Hence, in this research work, we have attempted to develop an Adaptive Buffer Management strategy to overcome congestion in a network with different traffic classes. Our ADQM strategy takes care of the GP, a refined one to take care of RT traffic class, and the notion of fair allocation, which depends on the connected load of a node at a given instant of time. Thus, the proposed ADQM strategy jointly takes care of the effect of priority and fair allocation. Our key contributions are as follows.

- The average queue size of a node is updated based on the proposed notion of the gradient of the difference of GP with respect to the differential queue size of two different layers. This is intended to link the priority with that of the queue size which inturn plays a vital role in controlling the output rates.
- The output rate of each node is obtained based on the modified GP and the average queue size, which is updated at each instant. Since the output rate is dependent on the modified GP and average queue size, the rate is controlled based on the updated queue size and hence will help in overcoming the congestion.
- The spare bandwidth gets updated based on the average queue size. The updated rate of a node is computed taking into consideration the connected load of a node. This ensures the fair distribution of the bandwidth depending upon the connected load and priority thus avoiding the issue of congestion.

In this paper, we have proposed a new congestion avoidance algorithm based on the adaptive queue size management that takes into account the GP and fair allocation of bandwidth. The average queue size is updated by the Exponentially Weighted Moving Average (EWMA) notion reinforced by the newly introduced notion of the gradient of the difference of GP with respect to differential queue size. The modified GP and the updated queue size at a given instant of time are used to compute the spare bandwidth. The updated bandwidth is fairly allocated among other nodes based on the connected load and the updated queue size. Thus, the proposed algorithm is based on the aspect of prioritizing the traffic class, fairness control, and queue management at individual nodes. QoS parameters have been compared with seven existing techniques and found to have improved results. We have evaluated the performance of the proposed algorithm using the NS3 platform [20] in a Linux environment under limited node mobility conditions. The nomenclatures used in this paper are provided in Table .

The rest of the paper is organized as follows. Section 2 provides related works and motivation while Section 3 presents the proposed rate control scheme. The concept of node priority is presented in Section 4. The weighted priority based fair queue gradient rate control scheme is described in Section 5 while Section 6 analyzes and discusses the experimental results and finally, Section 7 concludes the paper.

II. RELATED WORKS

In the early eighties, the issues of congestion control and avoidance were the research focus in the wireless paradigm, and in this regard in the year 1986, Jacobson and Karels have noticed the fact that bandwidth drops because of congestion collapses. This motivated Jacobson [21] to propose several new congestion avoidance algorithms followed by Floyd et al. [22]. Subsequently, a hop-by-hop control scheme is proposed by Mishra et al. [23] but their scheme does not take into account the TCP delay while this aspect is considered by Wan et al. [24] in their Congestion Detection and Avoidance Algorithm (CODA). Throughput and delay are the two key parameters of QoS in the network. In this regard, the issues of throughput instability and large delay in TCP are addressed by Zhai et al. [25] in multi-hop ad-hoc networks while Hull et al. [26] have addressed the issue of MAC layer protocol. Besides MAC layers, protocols in the transport layer have also been proposed in [27] to support multiple applications in the network. The congestion control algorithm is further improved by Sankarasubramaniam et al. [28] by proposing Event-to-Sink Reliable Transport (ESRT) scheme to detect reliable events in the network. The algorithms proposed in the above-mentioned research works don't work with all types of MAC protocols and also don't consider the notion of rate control. But the notion of rate control is considered by Alam et al. [29] while proposing a Congestion-aware and Rate-controlled Reliable Transport mechanism (CRRT) and by Liang et al. [18] for proposing a reliable transport protocol in WSN. Additionally, cross-layer optimization is also used to propose congestion control (DiffQ) algorithm [17].

In a wide variety of applications such as health monitoring, defence, environmental monitoring, and traffic surveillance, a huge amount of data are transmitted from source to sink. This leads to congestion which can be addressed using the notion of priority in the WSN. In this regard, Yaghmaee et al. [8] have proposed a priority based rate control mechanism in which the GP of each node is considered in a network. They have also extended their work [9] for different health care applications to monitor the health conditions of patients. Our earlier work [10] used the notion of traffic class priority to propose a priority-based adaptive rate control mechanism to control the congestion at each node. Priority concept is also used by different researchers [30]-[32] to improve the QoS of the network. It is found from the literature that using the notion of priority, QoS of the network could be improved but this could not overcome the issue of hotspots arising out of high data rate in the network. This problem is dealt by Monowar et al. [19] by proposing a single solution of the rate control scheme in WBAN.

It is known that unfairness in bandwidth distribution and queue overflow [33] are the major consequences of congestion. Therefore, the issues of fair distribution of bandwidth and queue management are addressed by researchers [33] to overcome congestion. Towards this end, Sarode *et al.* [33] have proposed three rate control mechanisms using buffer occupancy and priority traffic scheduler to achieve high reliability in long-distance transmission of packets. In [33], authors have not considered any queuing model at the node level. But the queue size of a node plays an important role in overcoming congestion. Queue size is considered in [34]–[36] to control congestion and inturn improve QoS. Rezaee et. al. [37] have used the concept of Active Queue Management (AQM) to propose an optimized congestion management protocol and a three-state machine to detect the congestion, and thereafter have extended their work [38] to health care applications. The notion of priority is also extended to wearable sensor based healthcare applications [39]. Despite using the notion of priority, there may be a possibility of heavy congestion during the transmission of medical data from a patient to the hospital server, that results in a huge amount of packet loss. To ameliorate this situation, a Relaxation Theory (RT) combined with Max-Min Fairness (MMF) mechanism has been proposed by Yaakob et al. [40] to increase the performance of the network. Recently, using the notion of priority queue, an IoT enabled WSN is also proposed by Chanak et al. [41] for smart healthcare applications.

The probability of congestion is higher near the sink node than that of other nodes due to the high flow rate of packets. To handle this issue, the notion of load balancing and fair bandwidth allocation along with the notion of priority have been considered by many researchers to avoid congestion. A scalable and distributed algorithm is proposed by Ee et al. [42] to control the congestion together with fairness for many-to-many routing in sensor networks. Subsequently, an Interference-aware Fair Rate Control (IFRC) scheme has been proposed by Rangwala et al. [12] to control the congestion by observing the average queue size in different topologies with multiple base stations. Further, to tackle the fairness issue, an Aggregate Fairness model with a localized Algorithm (AFA) has been proposed by Chen et al. [43] to fairly allocate the bandwidth in a network. The sole issue of fair distribution of the bandwidth is also addressed by Yin et al. [13] and Brahma et al. [14] to overcome congestion. In the context of fair distribution, Yin et al. [13] have proposed a rate-based Fair Aware Congestion Control (FACC) protocol to fairly allocate the bandwidth to different flows based on queue occupancy and hit frequency. But, Brahma et al. [14] have proposed a distributed congestion control to assign a fair and optimum transmission rate throughout the network. In these research works, authors have not considered the notion of priority together with the notion of fair allocation of bandwidth. It is to be noted that the joint use of fairness control and traffic class priority is expected to enhance the QoS effectively by overcoming congestion. In this regard, traffic class priority, fixed queue size at nodes of different layers of the network, and the connected load to a node are used in our previous work [16] to propose the a Priority based Fairness Rate Control (PFRC) scheme. The notion of fairness is also used to enhance the energy efficiency and network lifetime. A decentralized and Weighted Fairness guaranteed Congestion Control Protocol (WFCC) is proposed by Li et al. [44] and recently, a Reliable, Efficient, Fair and Interference-Aware Congestion Control (REFIACC)

algorithm has been proposed by Kafi *et al.* [15] to control the link interferences in each node of a large sensor network.

Besides the notion of fair bandwidth allocation, load balancing also plays an important role in overcoming congestion. This notion is used to propose traffic-balancing routing algorithm [45], QoS-aware Cross-layered Multichannel Routing (QCM2R) scheme [46], Congestion-aware Clustering and Routing (CCR) scheme [47], and Congestionaware and Traffic Load balancing Scheme (CTLS) [48] using multisink and multichannel approach. Often, the networks encounter random load conditions which is a challenging task. The load balancing task under such a scenario is also addressed by Ding et al. [49] to propose an optimizing Routing algorithm for congestion control. The issue of congestion in a grid sensor network is different from the conventional sensor network and this is addressed where the routing decisions are fixed using the grid densities and grid hop-count [50]. Additionally, QoS of the network is enhanced by Secure Selective Dropping Congestion Control (S2DCC) scheme by Tortelli et al. [51]. Further, the notion of hierarchy is used [52], [53] to distribute the load to overcome congestion. Hierarchy-based congestion control and the energy-balanced scheme are also proposed [54] to avoid congestion.

Very Recently, optimization algorithms such as particle swarm optimization [55], cuckoo search [56] algorithms, Imperialistic Competitive Algorithm (ICA) [57] are used to optimize the rate adjustment at the child nodes and the parent nodes. Zhuang *et al.* [58] have proposed a Congestion-Adaptive Data Collection (CADC) scheme to achieve accuracy for high data rate in cyber-physical system, and this algorithm uses the notion of lossy compression. Further to enhance the QoS, Tan *et al.* [59] have proposed a Differentiated Rate Control Data Collection (DRCDC) algorithm to avoid the congestion using information entropy theory.

Though several congestion control algorithms have been proposed, there is scope to improve upon the congestion strategies considering queue size, traffic class priority and fair distribution of bandwidth. Thus, we are motivated to pursue the same using the above criteria.

III. PROPOSED RATE CONTROL SCHEME

A. TRAFFIC CLASS PRIORITY

The RT and NRT traffic classes are prioritized in the proposed scheme. The NRT traffic class encompasses the high priority, medium priority, and low priority NRT classes and they are denoted as HNRT, MNRT, and LNRT respectively. Each of the traffic classes is assigned with different priority levels to differentiate the services. A node is categorized as either a parent node (*Pi*) or a child node (*Cj*). Hence, there are child nodes in a given j^{th} layer followed by the i^{th} layer of parent nodes. For the sake of illustration, a two-level hierarchy is shown in Fig. 1 but this can also be extended to multi-level hierarchy. There is a sink node followed by the parent nodes which are denoted as Q^{th} , $(Q + R)^{th}$, and $(Q + R + S)^{th}$

node. These nodes are designated as parent nodes which are connected to the child nodes ranging from $1, 2, ..., (Q+R+S-1)^{th}$. The proposed algorithm is developed for the general topology of Fig. 1 but for the sake of illustration, a specific topology is derived from the general topology with Q, R, and S values fixed at 5,3, and 2 respectively. This specific topology is shown in Fig. 2, which consists of one sink node, three parent nodes, and seven child nodes. In the network shown in Fig. 2, P1, P2, and P3 are the parent nodes and C1-C7 are the child nodes. The RT class is assigned to C1-C6, and P1 nodes, the HNRT class is assigned to C1, C2, C5, C7, and the LNRT class is assigned to C1, C3, C6, P1, and P2 nodes. We have tested our proposed algorithm on this topology but the algorithm is also valid for the general topology.



FIGURE 1. General topology.



FIGURE 2. Tree network topology.

Since four types of traffic classes are considered, we have defined the traffic class priority of each node as the sum of the source priorities. Then the proposed total priority for *i*th parent node and *j*th child node is defined as the sum of the product of traffic class priorities, geographical priorities, and the difference of the weighted RT class to the weighted sum of NRT classes as shown in (2) and (4). GP of each parent node is defined as the sum of the GPs of corresponding child nodes and the total priority of the parent node itself. Due to the unequal traffic at each child node, the transmission rate is also different for different nodes. So each node controls its transmission rate according to its GP i.e., $r_{out}^{Pi} \propto GP^{Pi}$. High

priority nodes will have higher transmission rates than the low priority nodes. To achieve both fair and efficient transmission rate of each node, we have considered both the notion of prioritizing the traffic class and the notion of fair distribution of the spare bandwidth at the parent level and the sink level in the network.

B. PROPOSED MODEL

The block diagram of the proposed scheme is shown in Fig. 3. The proposed scheme is validated on the tree topology shown in Fig. 2. As seen from Fig. 3, the Congestion Detection Unit (CDU) takes into account the difference of the output and the input rates of a given node to detect the onset of congestion. This information is fed to the Fairness Control Unit (FCU) which controls the rate to avoid the congestion. The Rate Adjustment Unit (RAU) takes input from the Fairness Control Unit (FCU) which gives rise to the fair



FIGURE 3. Block diagrammatic representation of the proposed scheme.

allocation of the bandwidth to different nodes. The FCU takes into account the adaptive queue size of a given node from the Adaptive Queue Management Unit (ADQMU) and the Priority Adjustment Unit (PAU). FCU determines the spare bandwidth and using the same it achieves a fair allocation of bandwidth to a given node. The queue size of each node is adapted taking care of the redefined GP and the current queue size of the nodes. Specifically, the average queue size is adapted using the proposed notion of the gradient of the difference of the weighted GP with respect to the differential current queue size. This adapts the average queue size of every node. This in turn, helps the FCU achieve the fair allocation of bandwidth. Finally, the RAU adapts the output rates of individual units and feeds to Congestion Control Unit (CCU) which in turn updates the rates of all the child nodes to avoid the congestion.

IV. NODE PRIORITY

Prioritizing traffic classes plays a vital role in circumventing congestion. Based on the traffic class priority, the Source Priority (SP) and Global Priority (GP) of a node can be determined. These GPs of the nodes do play an important role in congestion avoidance. In our proposed scheme, GP is taken into account in the Adaptive Queue Management strategy. Let T_s^{Sink} denote the service time of the current packet at the sink node and \overline{T}_s^{Sink} denotes the average service time. Let r_{out}^{Sink} denote output transmission rate of the sink node and r_{out}^{Pi} denotes the output transmission rate of i^{th} parent node. Let $\overline{Q}^{Sink}(n+1), \overline{Q}^{Pi}(n+1)$ and $\overline{Q}^{Cj}(n+1)$ denote the average queue sizes of the sink, parent and child nodes at n^{th} packet arrival. The traffic class priority (P_{TC}^{Pi}) of i^{th} parent nodes is defined as:

$$P_{TC}^{Pi} = \sum_{m} SP_{m}^{Pi}.$$
 (1)

where, *m* corresponds to the type of traffic class, $m \in \{RT, HNRT, MNRT, LNRT\}$ and SP_m^i is the source priority of the *i*th parent node. In order to prioritize each of the traffic classes, we have assigned weights to each of the traffic classes but maximum priority is assigned to the RT traffic classes which is by and large bursty in nature. With the above notion, using (1) the weighted total priority (P_w^{Pi}) of the *i*th parent node is proposed as:

$$P_{w}^{Pi} = P_{TC}^{Pi} \cdot P_{GE}^{Pi} + [W_{RT} - \gamma \cdot (W_{HNRT} + W_{MNRT} + W_{LNRT})].$$
(2)

where W_{RT} denotes the weight of RT traffic class, and W_{HNRT} , W_{MNRT} , and W_{LNRT} correspond to the weights of HNRT, MNRT, and LNRT traffic classes and the corresponding priority values assigned to them are 10, 6, 3, and 1 respectively, γ is a tunning parameter, $0 < \gamma \leq 1$ and P_{GE}^{Pi} is the geographical priority of i^{th} parent node and its value is different for different terrain such as hilly area or plain area but in our simulation this value is taken as 1 for plain area throughout the simulation. This indicates that more priority is assigned to the RT traffic class than those of NRT ones while computing P_w^{Pi} . Similarly, the traffic class priority (P_{TC}^{Cj}) of j^{th} child nodes is computed as:

$$P_{TC}^{Cj} = \sum_{m} SP_{m}^{Cj}.$$
 (3)

where, *m* corresponds to the set of traffic classes, $m \in \{RT, HNRT, MNRT, LNRT\}$ and SP_m^{Cj} is the source priority of the *j*th child nodes. Similarly, the weighted total priority (P_w^{Cj}) of *j*th child node using (3) is defined as:

$$P_{w}^{Cj} = P_{TC}^{Cj} \cdot P_{GE}^{Cj} + [W_{RT} - \gamma \cdot (W_{HNRT} + W_{MNRT} + W_{LNRT})].$$
(4)

where, γ is a tunning parameter, $0 < \gamma \leq 1$ and P_{GE}^{Cj} is the geographical priority of j^{th} child node and assigned with a value of 1. The weighted GP at the j^{th} child node (GP_w^{Cj}) using (4) is expressed as:

$$GP_w^{Cj} = P_w^{Cj}.$$
 (5)

The weighted GP at the i^{th} parent node (GP_w^{Pi}) using (5) and (2) is found to be:

$$GP_{w}^{Pi} = \sum_{j \in C(i)} GP_{w}^{Cj} + P_{w}^{Pi}.$$
 (6)

where, C(i) is the number of child nodes connected to i^{th} parent node. The weighted GP at the sink node (GP_w^{Sink}) using (6) is computed as:

$$GP_{w}^{Sink} = \sum_{i \in C(Sink)} GP_{w}^{Pi}.$$
(7)

where, C(Sink) is the number of child nodes connected to sink node. Thus, the GPs of the sink node, parent nodes, and the child nodes of the network are calculated.

In the literature [8], [9], the total priority of a node is contributed by P_{TC} and P_{GE} . But, in our case we have changed the notion of priority by assigning proper weights to the RT and NRT traffic classes. In our prioritization strategy, we have assigned maximum priority to the RT traffic class while other NRT traffic classes are assigned with reduced priority. The total priority of a node is contributed not only by the P_{TC} and P_{GE} but also by the weights of the RT class and weights of the NRT traffic classes. This additional priority contribution due to the RT class and NRT classes takes care of the connected load of a node in addition to its geographical priority. However, this notion which was used in our earlier work [11], [16] is also used in this proposed work.

V. WEIGHTED PRIORITY BASED FAIR QUEUE GRADIENT RATE CONTROL (WPFQGRC) SCHEME

Queue management is one of the pivots while developing the proposed algorithm. Particularly, Adaptive Queue management is imperative to handle the issue of congestion. Before developing the algorithm, we present briefly the notion of Adaptive Queue Management.

A. ADAPTIVE QUEUE MANAGEMENT (ADQM)

The Queue size of a sensor node plays an important role in handling congestion in WSN. Increase in number of packets in the queue implies that further increase of rate would lead to congestion at an early stage whereas less number of packets in the queue indicates that the rate can be further increased before the onset of congestion. Therefore, the fairness of the rate control mechanism is quite a pertinent issue where the spare bandwidth at a given instant of time needs to be distributed with the fairness.

The congestion detection parameters used in the literature are packet loss, queue size, packet service time, packet service ratio, node delay, channel busyness ratio, channel quality, and queue utilization. One of the key parameters for the indication of congestion is queue overflow. Therefore, management of the queue size is one of the crucial problems in wireless sensor networks [17]–[19], [22], [29]. Each node has an individual queue size that varies with the data transmission. When the queue size exceeds the maximum threshold limit, the output transmission rate is increased and vice versa. In this process, the queue management system reflects on the transmission rates of each node in the network.

In this regard, a fair rate control mechanism has been proposed [14] to achieve an optimum transmission rate with the notion of decoupling the congestion control unit and the fairness unit. In [14], a fixed value of queue size is considered throughout the network. But, each parent node in the network as shown in Fig. 2 is connected to a number of child nodes and hence a fixed queue size for all the nodes may not appropriate to take care of the congestion using the notion of fair distribution of the bandwidth. This problem can be overcome by adapting the queue size of each node depending on the traffic. In this regard, Alam *et al.* [29] have proposed a Congestion-aware and Rate-controlled Reliable Transport (CRRT) scheme where they have used the notion of EWMA to determine the average queue occupancy at a given node. Subsequently, Manowar *et al.* [19] have also used the same notion for hotspot avoidance in implantable body area networks. We have addressed this issue by improving the existing notion of AQM.

Therefore, it is imperative to develop a novel ADQM strategy taking into account the dominant features of the network. In this research work, we have taken care of the GP and the connected loads of a given node to develop a novel adaptive queue management strategy.

B. WPFQGRC STRATEGY

In order to adapt the queue size, it is necessary to consider the connected traffic of a node and its priority. Besides, due to the nature and rate of the traffic class, the traffic priorities need to be assigned with proper weightage. Therefore, the average queue size at a given instant of time at the sink, parent, and child nodes should take care of both the priority levels and connected load.

The queue size of a node is dependent on its connected load for congestion avoidance and its update also depends upon the priority and queue size of the destination node. For example, if the queue size of the parent node P1 needs to be updated, its adaptation strategy needs to take care of the current queue size and the priority of the destination node i.e. sink node besides considering the priority and the queue size of itself. Therefore, it is intitutively appealing that the adaptation of queue size will be contributed by the gradient of the difference of GPs of P1 and the sink node with that of queue sizes of P1 and the sink node. Similar notion is adapted for child nodes as well. This proposed notion is used to reinforce the existing EWMA [29] notion. Let $\overline{Q}^{Sink}, \overline{Q}^{Pi}$, and \overline{Q}^{Cj} denote the average queue size at sink node, i^{th} parent node and j^{th} child node respectively. Fig. 4 shows the average queue size of each node and the gradients allocated with different nodes. It is known that the average queue occupancy of a node with the notion of EWMA [29] is given by,

$$\overline{Q}(n+1) = (1-\alpha) \cdot \overline{Q}(n) + \alpha \cdot Q_{cur}, \qquad (8)$$

where, $\overline{Q}(n + 1)$ and $\overline{Q}(n)$ are the average queue sizes for $(n+1)^{th}$ and n^{th} packet respectively, Q_{cur} is the current queue size, and α is the tuning parameter between 0 and 1.

The queue size at $(n + 1)^{th}$ packet takes into account the current queue size and the average queue size at the previous packet. The increase in queue size necessitate the increase in



FIGURE 4. Tree topology with gradient concept.

rate to avoid congestion. In order to increase the transmission rate, the GP of a node needs to be changed accordingly. Since the GP of a node is affected by queue size, hence change in GP will affect the average queue size. Hence, as presented above, the change in GPs at two levels is expected to be affected by the change in the average queue size. Therefore, we modify (8) to propose the following adaptation of the average queue size.

$$\overline{Q}_{ADQM}(n+1) = (1-\alpha) \cdot \overline{Q}(n) + \alpha \cdot Q_{cur} + \delta \cdot Q_{cur} \cdot \left(\frac{\Delta GP_{w}(n)}{\Delta Q_{cur}(n)}\right).$$
(9)

where, $\left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)$ denotes the gradient of the differences, $\Delta GP_w(n) = GP_w(n) - GP_w(n-1)$ and $\Delta Q_{cur}(n) =$ $Q_{cur}(n) - Q_{cur}(n-1)$. Where $GP_w(n)$ and $GP_w(n-1)$ denote the weighted global priorities at n^{th} and $(n-1)^{th}$ packet respectively. Similarly, $Q_{cur}(n)$ and $Q_{cur}(n-1)$ denote the queue sizes at the n^{th} and $(n-1)^{th}$ packet arrivals respectively. In the topology shown in Fig. 4, the gradient is different for different connected nodes of either the sink node or the parent node. δ is a tunning parameter between 0 and 1. For example, the gradient at the sink will be evaluated based on the difference of GPs of the sink and the connected parent node with that of the average queue size. As seen in Fig. 4, there will be three gradients corresponding to three parent nodes. We need to compute the minimum average queue size at the sink node to overcome congestion. Therefore, we take into account the minimum of the three gradients. Therefore, (9) is modified with the proposed minimum gradient as,

$$Q_{ADQM}(n+1) = (1-\alpha) \cdot Q(n) + \alpha \cdot Q_{cur} + \delta \cdot Q_{cur} \cdot \left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)_{min}.$$
 (10)

where $\left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)_{min}$ denotes the minimum gradient between two nodes. The rate of change of gradient in (9) and (10) could often be more than unity and hence when multiplied by current value, will have large contribution to the queue adaptation. Therefore, the parameter, δ in (9) and (10) controls the queue adaptation rate. A large value of δ might lead to very large queue adaptation whereas very small value of δ will have very less contribution from the priority. In both the cases, performance of the network is affected and hence the parameter δ is tuned to obtain optimal throughput. Since, three parent nodes P1, P2 and P3 are connected to the sink node, the minimum gradient between sink node and *i*th parent node is computed as:

$$\left(\frac{\Delta GP_{w}(n)}{\Delta Q_{cur}(n)}\right)_{min}^{Sink,Pi} = \left(\frac{GP_{w}(n) - GP_{w}(n-1)}{Q_{cur}(n) - Q_{cur}(n-1)}\right)_{min}^{Sink,Pi}$$
(11)

which can also be expressed as:

$$\left\{\frac{GP_{w}(n) - GP_{w}(n-1)}{Q_{cur}(n) - Q_{cur}(n-1)}\right\}_{min}^{Sink,Pi}$$

$$= min \left\{\frac{\frac{GP_{w}^{Sink}(n) - GP_{w}^{P1}(n-1)}{Q_{cur}^{Sink}(n) - Q_{cur}^{P1}(n-1)}, \frac{GP_{w}^{Sink}(n) - GP_{w}^{P2}(n-1)}{Q_{cur}^{Sink}(n) - Q_{cur}^{P2}(n-1)}, \frac{GP_{w}^{Sink}(n) - GP_{w}^{P3}(n-1)}{Q_{cur}^{Sink}(n) - GP_{w}^{P3}(n-1)},$$
(12)

 $\left(\frac{Q_{w}^{Sink}(n) - Q_{w}^{P1}(n-1)}{Q_{cur}^{Sink}(n) - Q_{w}^{P1}(n-1)}\right), \quad \left(\frac{Q_{w}^{Sink}(n) - Q_{w}^{P2}(n-1)}{Q_{cur}^{Sink}(n) - Q_{cur}^{P1}(n-1)}\right), \quad \left(\frac{Q_{w}^{Sink}(n) - Q_{w}^{P2}(n-1)}{Q_{cur}^{Sink}(n) - Q_{w}^{P3}(n-1)}\right) \text{ and } \left(\frac{Q_{w}^{Sink}(n) - Q_{w}^{P3}(n-1)}{Q_{cur}^{Sink}(n) - Q_{w}^{P3}(n-1)}\right) \text{ are the gradients computed taking into account the sink node and the parent nodes.} \\ \left(\frac{Q_{w}^{Sink}(n) - Q_{w}^{P1}(n-1)}{Q_{cur}^{Sink}(n) - Q_{w}^{P1}(n-1)}\right) \text{ is defined as the gradient of weighted differential GP to the current differential queue size between the sink node and the parent node P1. Similarly, \\ \left(\frac{Q_{w}^{Sink}(n) - Q_{w}^{P2}(n-1)}{Q_{cur}^{Sink}(n) - Q_{w}^{P2}(n-1)}\right) \text{ and } \left(\frac{Q_{w}^{Sink}(n) - Q_{w}^{P3}(n-1)}{Q_{cur}^{Sink}(n) - Q_{w}^{P2}(n-1)}\right) \text{ correspond to the gradients while considering the sink node and the parent node and the parent nodes.}$

Next, we consider the parent nodes and the child nodes. Let $\left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)^{Pi,Cj}$ denotes the gradient of the difference of weighted GP of the *i*th parent node and the *j*th child node with that of the difference of the current queue sizes. It is to be noted that a given parent node is connected to many child nodes and therefore there will be different gradient values for different parent nodes. Analogous to the sink node case, we consider the minimum gradient $\left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)_{min}^{Pi,Cj}$ which corresponds to the minimum gradient between *i*th parent node and *j*th child node and in this regard, the minimum gradients between different parent nodes and child nodes are expressed as:

$$\left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)_{min}^{Pi,Cj} = \left(\frac{GP_w(n) - GP_w(n-1)}{Q_{cur}(n) - Q_{cur}(n-1)}\right)_{min}^{Pi,Cj}$$
(13)

Accordingly, the gradients are computed between the sink node, parent nodes and the child nodes. According to (10), the updated average queue size at the sink node using (12) is given by:

$$\overline{Q}_{ADQM}^{Sink}(n+1) = (1 - \alpha^{Sink}) \cdot \overline{Q}^{Sink}(n) + \alpha^{Sink} \cdot Q_{cur}^{Sink} + \delta^{Sink} \cdot Q_{cur}^{Sink} \cdot \left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)_{min}^{Sink,Pi}.$$
 (14)

Using the above minimum gradients (13) the average queue size of i^{th} parent node is updated as:

$$\overline{Q}_{ADQM}^{Pi}(n+1) = (1 - \alpha^{Pi}) \cdot \overline{Q}^{Pi}(n) + \alpha^{Pi} \cdot Q_{cur}^{Pi} + \delta^{Pi} \cdot Q_{cur}^{Pi} \cdot \left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)_{min}^{Pi,Cj}.$$
 (15)

where, α^{Pi} is a positive constant for i^{th} parent node and is between 0 and 1, δ^{Pi} is a gradient weight parameter between 0 and 1, Q_{cur}^{i} is the current queue size of the i^{th} parent node. Analogously $\overline{Q}_{ADQM}^{j}(n+1)$ for a child node is:

$$\overline{Q}_{ADQM}^{Cj}(n+1) = (1 - \alpha^{Cj}) \cdot \overline{Q}^{Cj}(n) + \alpha^{Cj} \cdot Q_{cur}^{Cj} + \delta^{Cj} \cdot Q_{cur}^{Cj} \cdot \left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)^{Pi,Cj}.$$
 (16)

where, α^{Cj} is a positive constant between 0 and 1. δ^{Cj} is a weighting parameter between 0 and 1 and $\left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)^{Pi,Cj}$ denotes the gradient between the *i*th parent node and the *j*th child node. The gradient of each child node is the only gradient computed using the connected parent node. Hence, the minimum gradient for a child node is the only available gradient as provided in the last term of (16).

It is to be noted that the output transmission rate at the sink node (r_{out}^{Sink}) is inversely proportional to the average service time $(\overline{T}_{s}^{Sink})$, i.e. $r_{out}^{Sink} \propto \frac{1}{\overline{T}_{s}^{Sink}}$, $r_{out}^{Sink} = k \cdot \frac{1}{\overline{T}_{s}^{Sink}}$, where k is a proportionality constant and its value is taken as unity. The \overline{T}_{s}^{Sink} is computed using the notion of Exponential Weighted Sum, which is defined as:

$$\overline{T}_{s}^{Sink}(n+1) = (1-\alpha_{1})\overline{T}_{s}^{Sink}(n) + \alpha_{1} \cdot T_{s}^{Sink}, \quad (17)$$

where, α_1 is a constant, $0 < \alpha_1 \le 1$.

As seen in Fig. 4, the parent nodes P1, P2 and P3 are connected to the sink node and hence the output rate of the sink node r_{out}^{Sink} is dependent on the weighted GP and the average queue size of the parent node and sink node as well. The queue size at the given node can be controlled to avoid congestion. If the queue size is less, then the output rate of the node can be increased further till the queue size is sufficient for the onset of congestion. The output rate of a given node is proportional to the weighted GP (GP_w) of that node and inversely proportional to the average queue size of the node. Hence for the sink node, the output rate using (7) and (14) can be expressed as:

$$r_{out}^{Sink}(n+1) \propto \frac{GP_{w}^{Sink}}{\overline{Q}_{ADOM}^{Sink}(n+1)}.$$
(18)

Analogously, for the i^{th} parent node the output rate is given by,

$$r_{out}^{Pi}(n+1) \propto \frac{GP_w^{Pi}}{\overline{\mathcal{Q}}_{ADOM}^{Pi}(n+1)}.$$
(19)

Taking the ratio of (18) and (19) we obtain,

$$\frac{r_{out}^{Pi}(n+1)}{r_{out}^{Sink}(n+1)} \propto \frac{GP_w^{Pi}}{\overline{Q}_{ADQM}^{Pi}(n+1)} \cdot \frac{\overline{Q}_{ADQM}^{Sink}(n+1)}{GP_w^{Sink}}.$$
 (20)

Hence, the output rate of the i^{th} parent node using (20) can be expressed as:

$$r_{out}^{Pi}(n+1) = k \cdot r_{out}^{Sink}(n+1) \cdot \frac{GP_{w}^{Pi}}{\overline{Q}_{ADQM}^{Pi}(n+1)} \cdot \frac{\overline{Q}_{ADQM}^{Sink}(n+1)}{GP_{w}^{Sink}}.$$
 (21)

where k is the proportionality constant. Selecting k to be unity, (21) reduces to,

$$r_{out}^{Pi}(n+1) = r_{out}^{Sink}(n+1) \cdot \frac{GP_{w}^{Pi}}{GP_{w}^{Sink}} \cdot \frac{\overline{Q}_{ADQM}^{Sink}(n+1)}{\overline{Q}_{ADQM}^{Pi}(n+1)}.$$
 (22)

In the same line as above, the output rate of a child node can be expressed as:

$$r_{out}^{Cj}(n+1) = r_{out}^{Pi}(n+1) \cdot \frac{GP_w^{Cj}}{GP_w^{Pi}} \cdot \frac{\overline{Q}_{ADQM}^{Pi}(n+1)}{\overline{Q}_{ADQM}^{Cj}(n+1)}.$$
 (23)

where, r_{out}^{j} is the output transmission rate of j^{th} child nodes. The parent nodes are transmitting data to the sink node and hence the output transmission rate of each parent node contributes to the input transmission rate of the sink node. Hence, the input transmission rate of the sink node (r_{in}^{Sink}) can be expressed as:

$$r_{in}^{Sink}(n+1) = \sum_{i \in C(Sink)} r_{out}^{Pi}(n+1).$$
 (24)

Similarly, the child nodes are transmitting data to their corresponding parent nodes. Hence the input transmission rate of the i^{th} parent node (r_{in}^{t}) is given by,

$$r_{in}^{Pi}(n+1) = \sum_{j \in C(i)} r_{out}^{Cj}(n+1).$$
(25)

where r_{out}^{j} is the output transmission rate of j^{th} child node. C(Sink) and C(i) denote the number of child nodes of the sink node and the number of child nodes of i^{th} parent node respectively.

It is known from the literature that Katabi *et al.* [34] have proposed the desired increase or decrease of the number of bytes of the aggregate traffic transmitted in a control interval. Thus, according to Katabi *et al.* [34] the aggregate feedback (ϕ), in terms of the number of bytes, is computed in the control interval and is given by,

$$\phi = \beta \cdot d \cdot S - \lambda \cdot Q. \tag{26}$$

where β and λ are tuning parameters used for stability, *d* is the average Round Trip Time (RTT), *S* is the defined as the difference of link rate and input traffic rate, and *Q* is the

persistent queue size. Dividing both sides of (26) by d we obtain,

$$\frac{\phi}{d} = \beta \cdot S - \lambda \cdot \frac{Q}{d}.$$
(27)

Since *d* is the time during which ϕ bytes are transmitted, the ratio $\frac{\phi}{d}$ can be perceived as the aggregate transmission rate (Δr). Before onset of congestion, the maximum output transmission rate (r_{out}) is equal to the link capacity. Therefore, the spare bandwidth (*S*) which is the difference of link capacity and the input traffic rate, can be expressed as ($r_{out} - r_{in}$). Besides, the queue size *Q* in (27) is the average queue obtained by the proposed ADQM strategy which is denoted by \overline{Q}_{ADQM} . Hence, the aggregate transmission rate for ADQM strategy (Δr_{ADQM}) can be expressed as:

$$\Delta r_{ADQM} = \beta \cdot (r_{out} - r_{in}) - \lambda \cdot (\frac{\overline{Q}_{ADQM}}{T_{cl}}).$$
(28)

where, β and λ are the stability parameters and are related as $\lambda = \beta^2 \sqrt{2}$ [34]. T_{cl} is the control interval [14], which is defined as:

$$T_{cl} = T_{Quedelay} + T_{Interpktarrtime} + 2 \cdot T_{MACpkttime}, \quad (29)$$

where, $T_{Quedelay}$ is the queueing delay, $T_{Interpktarrtime}$ is the inter-packet arrival time and $T_{MACpkttime}$ is the MAC packet transmission time. Hence, using (28) and (29) the aggregate transmission rates of sink and parent nodes can be computed as:

$$\Delta r_{ADQM}^{Sink}(n+1) = \beta \cdot (r_{out}^{Sink}(n+1) - r_{in}^{Sink}(n+1)) - \lambda \cdot \left(\frac{\overline{\mathcal{Q}}_{ADQM}^{Sink}(n+1)}{T_{cl}}\right), \quad (30)$$

$$\Delta r_{ADQM}^{Pi}(n+1) = \beta \cdot (r_{out}^{Pi}(n+1) - r_{in}^{Pi}(n+1)) - \lambda \cdot \left(\frac{\overline{\mathcal{Q}}_{ADQM}^{Pi}(n+1)}{T_{cl}}\right).$$
(31)

From (30) and (31), it may be noted that the distributions of spare bandwidths S^{Sink} or S^{Pi} are tantamount to the distribution of the aggregate transmission rate Δr_{ADQM}^{Sink} or Δr_{ADQM}^{Pi} . Therefore, the aggregate transmission rates (Δr_{ADQM}^{Sink}) and (Δr_{ADQM}^{Pi}) are distributed in a fair way among the connected loads of sink and parent nodes. In our earlier work [16], we have considered fair distribution among the nodes depending on the connected load of a given node. Adhering to the same notion, the updated transmission rate propagated by the sink node to the *i*th parent node, is given by;

$$= \begin{cases} r_{out,ADQM}^{P_{l}}(n+1) \\ r_{out}^{P_{l}}(n+1) + \frac{\Delta r_{ADQM}^{Sink}(n+1)}{L^{Sink}}, & \text{if } \Delta r_{ADQM}^{Sink}(n+1) > 0 \\ \frac{r_{out}^{P_{l}}(n+1)}{|\Delta r_{ADQM}^{Sink}(n+1)|}, & \text{if } \Delta r_{ADQM}^{Sink}(n+1) < 0. \end{cases}$$

$$(32)$$

where L^{Sink} is the connected load to the sink. Similarly, the new updated output transmission rate propagated by the i^{th} parent node towards the j^{th} child node after the fair distribution of the aggregate transmission rates among the respective loads (L^{Pi}) of i^{th} parent node, is given by;

$$= \begin{cases} r_{out,ADQM}^{CJ}(n+1) \\ r_{out}^{Cj}(n+1) + \frac{\Delta r_{ADQM}^{Pi}(n+1)}{L^{Pi}}, & \text{if } \Delta r_{ADQM}^{Pi}(n+1) > 0 \\ \frac{r_{out}^{Cj}(n+1)}{|\Delta r_{ADQM}^{Pi}(n+1)|}, & \text{if } \Delta r_{ADQM}^{Pi}(n+1) < 0. \end{cases}$$
(33)

The updated rates at the sink node, parent node, and child node are thus determined depending on the adaptive queue size of a given node. The flowchart of the WPFQGRC algorithm is presented in Fig. 5.

In the proposed scheme, the adaptive queue management depends on the gradient of change in GP of a given node with that of the change in queue size. Further, for a given parent node at *i*th level, the minimum gradient is considered among all the connected child nodes. Hence, the algorithm is valid for any number of child nodes connected to the parent node. Similarly, this gradient computation can be evaluated for any number of parent nodes at i^{th} level of the topology as shown in Fig. 4. Since the adaptation of queue size is dependent on the gradient computation at a node of i^{th} layer with that of the $(i-1)^{th}$ layer, adaptation at any number of layers can be computed. Thus, the algorithm is valid for any number of parent nodes in a layer, any number of connected child nodes and finally for any number of layers. Hence, the algorithm is suitable for a generalized network topology as shown in Fig. 1.

VI. RESULTS AND DISCUSSIONS

The network topology shown in Fig. 2 is considered in our simulation. This particular topology is derived from the general topology [8] shown in Fig. 1. The different traffic classes considered in our simulation are the RT and different NRT traffic classes. Specifically, three different NRT traffic classes such as HNRT, MNRT, and LNRT traffic classes are considered. This specific topology with the proposed algorithm is implemented in NS3 [20] platform in Linux environment. We have developed the necessary codes for the algorithms implemented in NS3 platform. Different traffic classes of the nodes are also shown in Fig. 2 where the communication among different nodes is established by AODV protocol. It is to be noted that RT data needs to be assigned with the highest priority followed by HNRT, MNRT, and LNRT data. Therefore, in our simulation, RT data is assigned with a priority value of 10, whereas HNRT, MNRT, and LNRT are assigned with priority values of 6, 3, and 1 respectively. Other simulation parameters are presented in Table 1.

The notions of priority, fair distribution of bandwidth and queue size are taken into consideration while developing the proposed algorithm. Therefore, the fair distribution of the Algorithm 1 Weighted Priority Based Fair Queue Gradient Rate Control for Congestion Avoidance

Input: Initialize the parameters: T_s^{Sink} , $t_{simulation}$, γ , α , α_1 , α^{Sink} , α^{Pi} , α^{Cj} , δ^{Sink} , δ^{Pi} , δ^{Cj} , Q_{cur}^{Sink} , Q_{cur}^{Cj} , Q_{cur}^{Cj} , β , λ , k, and traffic class priorities.

Output: Updated output transmission rates of parent nodes and child nodes.

- 1: Initialize algorithm parameters
- 2: for n=0 to $t_{simulation}$ do
- $\overline{T}_{s}^{Sink}(n+1) = (1-\alpha_{1})\overline{T}_{s}^{Sink}(n) + \alpha_{1} \cdot T_{s}^{Sink}$ //Average service time calculation for Sink node $r_{out}^{Sink}(n+1) = k \cdot \frac{1}{\overline{T}_{s}^{Sink}(n+1)}$ //Output
- transmission rate calculation for Sink node $\overline{Q}_{ADQM}^{Sink}(n+1) = (1 \alpha^{Sink}) \cdot \overline{Q}^{Sink}(n) + \alpha^{Sink} \cdot Q_{cur}^{Sink} + \delta^{Sink} \cdot Q_{cur}^{Sink} \cdot \left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)_{min}^{Sink,Pi}$ 5: //Calculation of average queue size of Sink node

based on ADQM strategy

6:
$$\overline{Q}_{ADQM}^{Pi}(n+1) = (1 - \alpha^{Pi}) \cdot \overline{Q}^{Pi}(n) + \alpha^{Pi} \cdot Q_{cur}^{Pi} + \delta^{Pi} \cdot Q_{cur}^{Pi} + \delta^{Pi} \cdot Q_{cur}^{Pi} \cdot \left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)_{min}^{Pi,Cj}$$

//Calculation of average queue size of parent nodes based on ADQM strategy

7: $\overline{Q}_{ADQM}^{Cj}(n+1) = (1 - \alpha^{Cj}) \cdot \overline{Q}^{Cj}(n) + \alpha^{Cj} \cdot Q_{cur}^{Cj} + \delta^{Cj} \cdot Q_{cur}^{Cj} + \delta^{Cj} \cdot Q_{cur}^{Cj} \cdot \left(\frac{\Delta GP_w(n)}{\Delta Q_{cur}(n)}\right)^{Pi,Cj}$

//Calculation of average queue size of child nodes based on ADQM strategy _____Sin

8:
$$r_{out}^{Pi}(n+1) = r_{out}^{Sink}(n+1) \cdot \frac{GP_w^{Pi}}{GP_w^{Sink}} \cdot \frac{Q_{ADQM}(n+1)}{\overline{Q}_{ADQM}^{Pi}(n+1)}$$

//Calculation of output transmission rate of parent

nt nodes based on ADQM strategy

$$r_{out}^{Cj}(n+1) = r_{out}^{Pi}(n+1) \cdot \frac{GP_w^{Cj}}{GP_w^{Pi}} \cdot \frac{\overline{\mathcal{Q}}_{ADQM}^{Pi}(n+1)}{\overline{\mathcal{Q}}_{ADOM}^{Cj}(n+1)}$$

//Calculation of output transmission rate of child nodes based on ADQM strategy

- $r_{in}^{Sink}(n+1) = \sum_{i \in C(Sink)} r_{out}^{Pi}(n+1)$ //Calculation of input transmission rate of Sink 10: node based on ADQM strategy
- $r_{in}^{Pi}(n+1) = \sum_{j \in C(i)} r_{out}^{Cj}(n+1)$ 11: //Calculation of input transmission rate of parent nodes based on ADQM strategy 12:
- $\Delta r_{ADQM}^{Sink}(n+1) = \beta \cdot (r_{out}^{Sink}(n+1) r_{in}^{Sink}(n+1)) \lambda \cdot \left(\frac{\overline{Q}_{ADQM}^{Sink}(n+1)}{T_{cl}}\right) //Compute the aggregate transmission$

rate at the sink node if $\Delta r_{ADOM}^{Sink}(n+1) > 0$ then 13:

14:

9:

$$r_{out}^{Pi}(n+1) = r_{out}^{Pi}(n+1) + \frac{\Delta r_{ADQM}^{Sink}(n+1)}{L^{Sink}}$$

//Evaluate the updated output transmission rate of parent node else

15: $r_{out}^{Pi}(n+1) = \frac{r_{out}^{Pi}(n+1)}{|\Delta r_{ADQM}^{Sink}(n+1)|}$ end if 16: $\Delta r_{ADQM}^{Pi}(n+1) = \beta \cdot (r_{out}^{Pi}(n+1) - r_{in}^{Pi}(n+1)) - \lambda \cdot \left(\frac{\overline{Q}_{ADQM}^{Pi}(n+1)}{T_{cl}}\right) //\text{Compute the aggregate}$ transmission rate at the *i*th parent node 17: if $\Delta r_{ADQM}^{Pi}(n+1) > 0$ then 18: $r_{out}^{Cj}(n+1) = r_{out}^{Cj}(n+1) + \frac{\Delta r_{ADQM}^{Pi}(n+1)}{L^{Pi}} //\text{Evaluation of the updated output transmission}$ rate of *j*th child node else 19: $r_{out}^{Cj}(n+1) = \frac{r_{out}^{Cj}(n+1)}{L^{Pi}} = \frac{r_{out}^{Cj}(n+1)}{|\Delta r_{ADQM}^{Pi}(n+1)|}$ end if end for

20: Return the updated output rates to child nodes

bandwidth is affected by the priorities of the traffic classes and the queue size as well. Though the performance of each node is analyzed, for the sake of illustration, we have presented the results of the child nodes C1 and C2 and the corresponding parent node P1.

It is to be noted that the adaptation of queue size for each node is imperative for congestion avoidance and hence three QoS parameters such as throughput, packet loss, and packet delay are determined for each node. The above three parameters for child node C1 are presented in Figs. 6, 7, and 8. In order to demonstrate individual algorithm's effect on throughput, we have plotted throughput of different algorithms separately in Figs. 6. It may be observed from Fig. 6 that the throughput for the proposed WPFQGRC algorithm is the highest where as it is lowest for Brahma et al.'s algorithm. Similarly, the variations of three parameters for child node C4 are shown in Figs. 9, 10 and 11 and the corresponding results for the parent node P1 are presented in Figs. 12, 13 and 14. As seen in Fig. 2, the RT load and three types of NRT loads such as HNRT, MNRT, and LNRT are connected to child node C1 and therefore the effect of these connected loads are expected to be evident in the parent node P1. In Fig. 6, the lowest throughput of Brahma et al.'s algorithm is because of the fact that the notions of priority and the queue size adaptation are absent. But, the throughputs of the Monowar et al., Sarode et al., PFRC, DDRC, and WPDDRC algorithms are found to be more than that of the Brahma et al.'s algorithm but less than that of the proposed WPFQGRC algorithm. This is because of the combined strategy adopted by WPFQGRC algorithm. In case of packet loss, similar observations are also made in Fig. 7. It is also observed from Fig. 7 that minimum loss occurs in WPFQGRC algorithm but the maximum packet loss occurs for Monowar et al.'s algorithm as the notion of fair bandwidth distribution is absent in Monowar et al.'s algorithm.



FIGURE 5. Flowchart of proposed WPFQGRC algorithm.

From the implementation point of view, we have only considered the notion of node priority and the queue management strategy for Monowar et al.'s algorithm. The concept of hotspot avoidance is not considered in our implementation. The delay parameter for the corresponding node is shown in Fig. 8. It is evident from Fig. 8 that there is minimum amount of delay for WPFQGRC algorithm. This may be attributed to the fact that the proposed algorithm is based on the combined notion of traffic class priority, fair distribution of bandwidth, and ADQM strategy. In addition to the above three aspects, the algorithm takes into account the number of child nodes of a given parent node. In case of child node C4, similar effects are also observed in Fig. 9, 10 and 11. The rates of the child nodes also affect the parent node and in this regard, node P1 has two child nodes and hence the rate of parent node P1 is expected to be higher than either of the nodes. This phenomenon is reflected in Fig. 12. Therefore, as seen from Fig. 12, the proposed algorithm produced throughput of 950 Kbps while other algorithms produced less throughputs than that of the proposed algorithm. As observed from Fig. 13 and Fig. 14, the findings of the loss and delay

parameters of parent node P1 are similar to those of child nodes C1 and C4.

Parameter	Values		
Simulation platform, OS	NS-3, Linux		
Simulation area	$100 * 100 \ m^2$		
MAC layer	IEEE 802.11		
Number of sensor nodes	11		
Packet size	172 Bytes		
Data rate	2 Mbps		
Tx power	281.84 mW		
Operating frequency	5 GHz		
Traffic source	CBR		
Routing protocol	AODV		
Mobility model	Random Walk2d Mobility Model		
Node C2 mobility	10 m/s		
Pause time	No pause time		
Simulation time	100 Seconds		

TABLE 1. Simulation parameters.

A. EFFECT OF TRAFFIC CLASS PATTERNS, NODE MOBILITY, AND AVERAGE QUEUE SIZE

1) EFFECT OF TRAFFIC CLASS PATTERNS

Since Brahma et al.'s algorithm does not take into account the priority of traffic classes, the algorithm is excluded from the study of the effect of the traffic patterns. In this regard, different traffic patterns are connected to child nodes C1, C2, C3, and C4 and their effects on the throughputs of these four child nodes are studied. Further, the effect on the corresponding parent node P1 is also analyzed. Though any type of traffic pattern can be connected to a node, for the sake of illustration, three different traffic patterns namely TP1, TP2, and TP3 are considered at child node C2. These patterns are TP1 (RT+HNRT+MNRT), TP2 (RT+HNRT), and TP3 (RT+MNRT+LNRT). Out of these three traffic patterns, TP1, and TP2 include the RT traffic class and are expected to have more priorities than that of TP3, which does not include the RT traffic class. Fig. 15(a) and (b) show the throughputs of three different traffic patterns for seven different algorithms. As observed in Fig. 15(a) and (b), the throughputs for the three traffic patterns of Monowar et al.'s algorithm are the lowest while the throughputs of the proposed WPFQGRC algorithm are the highest ones. This is due to the fact that the proposed WPFQGRC algorithm takes care of the priorities of traffic class pattern, while simultaneously adapting the queue size and distributing the spare bandwidth in a fair manner. As TP3 pattern has low priority due to the absence of RT class, the corresponding throughput is the lowest. Similar observations are also made for Yaghmaee et al., DDRC, Sarode et al., WPDDRC, and PFRC algorithms. But, the throughput is found to be maximum for the TP1 traffic



FIGURE 6. Node C1: Comparison of throughput of different algorithms.



FIGURE 7. Node C1: Comparison of packet loss.



FIGURE 8. Node C1: Comparison of packet delay.

pattern, which includes RT traffic class that has been assigned with high priority. Hence, it is found that throughputs are dependent on the traffic patterns and particularly, as expected, high throughput is achieved for high priority traffic pattern.

As C2 is connected to P1, the effect of traffic patterns on the parent node P1 is also studied which is shown in Fig. 16(a) and (b). Though the traffic patterns connected to



FIGURE 9. Node C4: Comparison of throughput of different algorithms.



FIGURE 10. Node C4: Comparison of packet loss.



FIGURE 11. Node C4: Comparison of packet delay.

child nodes may vary, for the sake of illustration, all the child nodes are connected to preassigned traffic patterns. The node P1 is also connected with a fixed traffic pattern. Therefore, the throughput of P1 depends upon the combined effect of the traffic patterns of the child nodes and the parent node as well. In this case, as observed from Fig. 16, the throughputs achieved by Monowar *et al.*'s algorithm are the lowest



FIGURE 12. Node P1: Comparison of throughput of different algorithms.



FIGURE 13. Node P1: Comparison of packet loss.



FIGURE 14. Node P1: Comparison of packet delay.

while the throughputs achieved by the proposed WPFQGRC are the highest ones. Further, in case of Monowar *et al.*'s algorithm, TP3 traffic pattern produced the lowest throughput while TP1 produced the highest one. This is because of the fact that TP1 has higher traffic class priority than that of TP3 pattern. Similar observations are also made for the other algorithms. In case of the parent node also, the performance



FIGURE 15. Node C2: Different traffic patterns (TP1(RT+HNRT+MNRT), TP2(RT+HNRT), and TP3(RT+MNRT+LNRT)) and their effect on throughput.

of the proposed WPFQGRC algorithm is the best among all the algorithms.

2) EFFECT OF NODE MOBILITY

The effect of limited node mobility of a given node on the throughput is also studied. For the sake of illustration, node C2 is allowed to move in the neighborhood of its original position i.e. near the neighbor nodes, C1 and C3 with a mobility rate of 10 metre/sec. The throughputs of C2, with and without mobility, for all the algorithms are presented in Fig. 17. As observed in Fig. 17(a), because of the mobility of a given node, the throughput is not stable and hence starts from a low value and thereafter rises to a high value over some time interval and as the mobility is restricted to a limited zone, the throughput finally settles down to a particular value. The node moves randomly with a constant velocity i.e. 10 metre/sec. Hence, initially because of the change of positions, the GP is affected and hence the queue adaptation is affected which inturn increases the throughput. After sometime, though there is movement, the average of GP of different positions is constant and hence the throughput remains constant after sometime. It is also observed that the



FIGURE 16. Node P1: Different traffic patterns (TP1(RT+HNRT+MNRT), TP2(RT+HNRT), and TP3(RT+MNRT+LNRT)) and their effect on throughput.

throughputs obtained for mobile node cases are close to that of the static cases. Similar findings are also observed for the other four algorithms as shown in Fig. 17(b). As observed in Fig. 17(b), in some cases, the throughputs for the mobile nodes rise up to 10 *sec* and thereafter settle at some values. The throughput obtained for the proposed WPFQGRC algorithm is the highest for both static as well as mobile cases. Hence, it is inferred that the mobility of a node affects the throughput during the initial period but subsequently the throughput is found to be close to that of the static case.

3) EFFECT OF TRAFFIC PATTERN ON QUEUE SIZE

To overcome congestion, queue size of a node changes depending on its connected load. The average queue sizes of different child nodes and the corresponding parent nodes over the simulation time are shown in Fig. 18, 19, and 20. Fig. 18 shows the queue sizes for the four child nodes namely C1-C4 and the corresponding parent node P1. As observed in Fig. 18, at a given instant of time, the queue size of the parent node P1 is less than those of the child nodes. This is because of the fact that the rate of P1 is higher than those of the child nodes and hence the queue size is expected



(b)

FIGURE 17. Node C2: Effect of mobility on the throughput of node-C2.



FIGURE 18. Average queue size of node-P1 and corresponding child nodes.

to be lower. Similar observations are made in Fig. 19 for parent node P2 and the corresponding child nodes. Analogous phenomenon is also observed in Fig. 20. In case of child node C2, the variations of average queue size due to different traffic patterns TP1, TP2, and TP3 are shown in Fig. 21. It is also known that the traffic patterns affect the GP of node C2 which in turn affects its average queue size. Hence,



FIGURE 19. Average queue size of node-P2 and corresponding child nodes.



FIGURE 20. Average queue size of node-P3 and corresponding child node.



FIGURE 21. Average queue size of node-C2 for different traffic patterns, (TP1(RT+HNRT+MNRT), TP2(RT+HNRT), and TP3(RT+MNRT+LNRT)).

the average queue size is different for different traffic pattern as shown in Fig. 21. The traffic pattern TP1 has high priority and therefore demands a high transmission rate to avoid congestion. Hence, the queue size is also large to overcome congestion. Similarly, the traffic pattern TP3 has the lowest priority and hence average queue size is small, which is evident from Fig. 21. Hence, it is inferred that traffic patterns do affect the queue size which is expected to overcome the congestion.

VII. CONCLUSION

It is known that considering either the notion of fairness or priority does not overcome congestion completely. This motivated us to develop the proposed novel WPFQGRC rate control scheme that takes care of the fair distribution of the spare bandwidth, the priority of traffic classes, and the proposed Adaptive Queue Management. The proposed ADQM strategy depends on the gradient of the differential GP with respect to a differential queue size corresponding to different levels of the network. The QoS parameters such as throughput, loss, and delay are found to improve as compared to other algorithms. It is also found that the proposed algorithm produced the highest throughput while overcoming congestion. The effect of traffic class pattern is also investigated and it is found that the traffic pattern affects the throughput. It is observed that high priority traffic class leads to high throughput and vice versa. The effect of limited node mobility is also investigated and it is found that because of the limited mobility, there is a drop in the throughput initially but thereafter it increased to a value close to the static case. It is to be noted that the limited mobility of the node does not deteriorate the performance appreciably. The effect of different traffic patterns on queue size is also studied and it is observed that the traffic pattern does affect the queue size. Future work will focus on the study of the effect of the mobile sink node, number of child nodes and the parent nodes on throughput and a modified queue management strategy. Instead of a single tree topology, number of clusters with cluster heads may be considered in future which can be implemented in a real time testbed.

ACKNOWLEDGMENT

The research facility provided by Siksha 'O' Anusandhan (Deemed to be University), Bhubaneswar is greatly acknowledged for pursuing this research work.

REFERENCES

- I. F. Akyildiz, W. Su, Y. Sankarasubramaniam, and E. Cayirci, "Wireless sensor networks: A survey," *Comput. Netw.*, vol. 38, no. 4, pp. 393–422, 2002.
- [2] I. F. Akyildiz, T. Melodia, and K. R. Chowdhury, "A survey on wireless multimedia sensor networks," *Comput. Netw.*, vol. 51, no. 4, pp. 921–960, Mar. 2007.
- [3] J. Yick, B. Mukherjee, and D. Ghosal, "Wireless sensor network survey," *Comput. Netw.*, vol. 52, no. 12, pp. 2292–2330, Aug. 2008.
- [4] C. Wang, K. Sohraby, B. Li, M. Daneshmand, and Y. Hu, "A survey of transport protocols for wireless sensor networks," *IEEE Netw.*, vol. 20, no. 3, pp. 34–40, May 2006.
- [5] M. A. Kafi, D. Djenouri, J. B. Othman, and N. Badache, "Congestion control protocols in wireless sensor networks: A survey," *IEEE Commun. Surveys Tuts.*, vol. 16, no. 3, pp. 1369–1390, 3rd Quart., 2014.
- [6] A. Ghaffari, "Congestion control mechanisms in wireless sensor networks: A survey," J. Netw. Comput. Appl., vol. 52, pp. 101–115, Jun. 2015.
- [7] D. Pandey and V. Kushwaha, "An exploratory study of congestion control techniques in wireless sensor networks," *Comput. Commun.*, vol. 157, pp. 257–283, May 2020.
- [8] M. H. Yaghmaee and D. Adjeroh, "Priority-based rate control for service differentiation and congestion control in wireless multimedia sensor networks," *Comput. Netw.*, vol. 53, no. 11, pp. 1798–1811, 2009.

- [9] M. H. Yaghmaee, N. F. Bahalgardi, and D. Adjeroh, "A prioritization based congestion control protocol for healthcare monitoring application in wireless sensor networks," *Wireless Pers. Commun.*, vol. 72, no. 4, pp. 2605–2631, 2013.
- [10] S. K. Swain and P. K. Nanda, "Priority-based adaptive rate control for wireless sensor networks," in *Proc. IEEE Int. Advance Comput. Conf.* (*IACC*), Feb. 2014, pp. 370–374.
- [11] S. K. Swain and P. K. Nanda, "Priority based adaptive rate control in wireless sensor networks: A difference of differential approach," *IEEE Access*, vol. 7, pp. 112435–112447, 2019.
- [12] S. Rangwala, R. Gummadi, R. Govindan, and K. Psounis, "Interferenceaware fair rate control in wireless sensor networks," ACM SIGCOMM Comput. Commun. Rev., vol. 36, no. 4, pp. 63–74, Aug. 2006.
- [13] X. Yin, X. Zhou, R. Huang, Y. Fang, and S. Li, "A fairness-aware congestion control scheme in wireless sensor networks," *IEEE Trans. Veh. Technol.*, vol. 58, no. 9, pp. 5225–5234, Nov. 2009.
- [14] S. Brahma, M. Chatterjee, K. Kwiat, and P. K. Varshney, "Traffic management in wireless sensor networks: Decoupling congestion control and fairness," *Comput. Commun.*, vol. 35, no. 6, pp. 670–681, 2012.
- [15] M. A. Kafi, J. Ben-Othman, A. Ouadjaout, M. Bagaa, and N. Badache, "REFIACC: Reliable, efficient, fair and interference-aware congestion control protocol for wireless sensor networks," *Comput. Commun.*, vol. 101, pp. 1–11, Mar. 2017.
- [16] S. K. Swain and P. K. Nanda, "Priority based fairness rate control in wireless sensor networks," in *Proc. Int. Conf. Wireless Commun., Signal Process. Netw. (WiSPNET)*, Mar. 2017, pp. 2206–2210.
- [17] A. Warrier, S. Janakiraman, S. Ha, and I. Rhee, "DiffQ: Practical differential backlog congestion control for wireless networks," in *Proc. IEEE 28th Conf. Comput. Commun. (INFOCOM)*, Apr. 2009, pp. 262–270.
- [18] L. Liang, D. Gao, and V. C. M. Leung, "Queue-based congestion detection and multistage rate control in event-driven wireless sensor networks," *Wireless Commun. Mobile Comput.*, vol. 14, no. 8, pp. 818–830, Jun. 2014.
- [19] M. M. Monowar and F. Bajaber, "Towards differentiated rate control for congestion and hotspot avoidance in implantable wireless body area networks," *IEEE Access*, vol. 5, pp. 10209–10221, 2017.
- [20] NS3: Network Simulator. Accessed: Mar. 2015. [Online]. Available: https://www.nsnam.org
- [21] V. Jacobson, "Congestion avoidance and control," ACM SIGCOMM Comput. Commun. Rev., vol. 18, no. 4, pp. 314–329, 1988.
- [22] S. Floyd and V. Jacobson, "Random early detection gateways for congestion avoidance," *IEEE/ACM Trans. Netw.*, vol. 1, no. 4, pp. 397–413, Aug. 1993.
- [23] P. P. Mishra, H. Kanakia, and S. K. Tripathi, "On hop-by-hop rate-based congestion control," *IEEE/ACM Trans. Netw.*, vol. 4, no. 2, pp. 224–239, Apr. 1996.
- [24] C.-Y. Wan, S. B. Eisenman, and A. T. Campbell, "CODA: Congestion detection and avoidance in sensor networks," in *Proc. 1st Int. Conf. Embedded Netw. Sensor Syst.*, 2003, pp. 266–279.
- [25] H. Zhai, X. Chen, and Y. Fang, "Improving transport layer performance in multihop ad hoc networks by exploiting MAC layer information," *IEEE Trans. Wireless Commun.*, vol. 6, no. 5, pp. 1692–1701, May 2007.
- [26] B. Hull, K. Jamieson, and H. Balakrishnan, "Mitigating congestion in wireless sensor networks," in *Proc. 2nd Int. Conf. Embedded Netw. Sensor Syst. (SenSys)*, 2004, pp. 134–147.
- [27] Y. G. Iyer, S. Gandham, and S. Venkatesan, "STCP: A generic transport layer protocol for wireless sensor networks," in *Proc. 14th Int. Conf. Comput. Commun. Netw. (ICCCN)*, 2005, pp. 449–454.
- [28] Y. Sankarasubramaniam, O. B. Akan, and I. F. Akyildiz, "ESRT: Eventto-sink reliable transport in wireless sensor networks," in *Proc. 4th ACM Int. Symp. Mobile Ad Hoc Netw. Comput.*, 2003, pp. 177–188.
- [29] M. M. Alam and C. S. Hong, "CRRT: Congestion-aware and ratecontrolled reliable transport in wireless sensor networks," *IEICE Trans. Commun.*, vol. 92, no. 1, pp. 184–199, 2009.
- [30] R. Sumathi and R. Srinivasan, "QoS aware routing protocol to improve reliability for prioritised heterogeneous traffic in wireless sensor network," *Int. J. Parallel, Emergent Distrib. Syst.*, vol. 27, no. 2, pp. 143–168, 2012.
- [31] C. Wang, B. Li, K. Sohraby, M. Daneshmand, and Y. Hu, "Upstream congestion control in wireless sensor networks through cross-layer optimization," *IEEE J. Sel. Areas Commun.*, vol. 25, no. 4, pp. 786–795, May 2007.
- [32] L. Tshiningayamwe, G.-A. Lusilao-Zodi, and M. E. Dlodlo, "A priority rate-based routing protocol for wireless multimedia sensor networks," in *Advances in Nature and Biologically Inspired Computing*, N. Pillay, A. P. Engelbrecht, A. Abraham, M. C. du Plessis, V. Snášel, and A. K. Muda, Eds. Cham, Switzerland: Springer, 2016, pp. 347–358.

- [33] S. S. Sarode and J. W. Bakal, "A data transmission protocol for wireless sensor networks: A priority approach," *J. Telecommun., Electron. Comput. Eng.*, vol. 10, no. 3, pp. 65–73, 2018.
- [34] D. Katabi, M. Handley, and C. Rohrs, "Congestion control for high bandwidth-delay product networks," in *Proc. Conf. Appl., Technol., Architectures, Protocols Comput. Commun. (SIGCOMM)*, 2002, pp. 89–102.
- [35] S. M. Aghdam, M. Khansari, H. R. Rabiee, and M. Salehi, "WCCP: A congestion control protocol for wireless multimedia communication in sensor networks," *Ad Hoc Netw.*, vol. 13, pp. 516–534, Feb. 2014.
 [36] X. Zhang, D. Li, and Y. Zhang, "Maximum throughput under admission
- [36] X. Zhang, D. Li, and Y. Zhang, "Maximum throughput under admission control with unknown queue-length in wireless sensor networks," *IEEE Sensors J.*, vol. 20, no. 19, pp. 11387–11399, Oct. 2020.
- [37] A. Rezaee, M. Yaghmaee, and A. Rahmani, "COCM: Class based optimized congestion management protocol for healthcare wireless sensor networks," *Wireless Sensor Netw.*, vol. 5, no. 7, pp. 137–149, 2013.
- [38] A. A. Rezaee, M. H. Yaghmaee, and A. M. Rahmani, "Optimized congestion management protocol for healthcare wireless sensor networks," *Wireless Pers. Commun.*, vol. 75, no. 1, pp. 11–34, Mar. 2014.
- [39] A. Alsiddiky, W. Awwad, H. Fouad, A. S. Hassanein, and A. M. Soliman, "Priority-based data transmission using selective decision modes in wearable sensor based healthcare applications," *Comput. Commun.*, vol. 160, pp. 43–51, Jul. 2020.
- [40] N. Yaakob and I. Khalil, "A novel congestion avoidance technique for simultaneous real-time medical data transmission," *IEEE J. Biomed. Health Informat.*, vol. 20, no. 2, pp. 669–681, Mar. 2016.
- [41] P. Chanak and I. Banerjee, "Congestion free routing mechanism for IoTenabled wireless sensor networks for smart healthcare applications," *IEEE Trans. Consum. Electron.*, vol. 66, no. 3, pp. 223–232, Aug. 2020.
- [42] C. T. Ee and R. Bajcsy, "Congestion control and fairness for many-toone routing in sensor networks," in *Proc. 2nd Int. Conf. Embedded Netw. Sensor Syst. (SenSys)*, 2004, pp. 148–161.
- [43] S. Chen and Z. Zhang, "Localized algorithm for aggregate fairness in wireless sensor networks," in *Proc. 12th Annu. Int. Conf. Mobile Comput. Netw. (MobiCom)*, 2006, pp. 274–285.
- [44] G. Li, J. Li, and B. Yu, "Lower bound of weighted fairness guaranteed congestion control protocol for WSNs," in *Proc. IEEE INFOCOM*, Mar. 2012, pp. 3046–3050.
- [45] D. D. Tan and D.-S. Kim, "Dynamic traffic-aware routing algorithm for multi-sink wireless sensor networks," *Wireless Netw.*, vol. 20, no. 6, pp. 1239–1250, 2014.
- [46] W. Rehan, S. Fischer, M. Rehan, Y. Mawad, and S. Saleem, "QCM2R: A QoS-aware cross-layered multichannel multisink routing protocol for stream based wireless sensor networks," *J. Netw. Comput. Appl.*, vol. 156, Apr. 2020, Art. no. 102552.
- [47] M. Farsi, M. Badawy, M. Moustafa, H. Arafat Ali, and Y. Abdulazeem, "A congestion-aware clustering and routing (CCR) protocol for mitigating congestion in WSN," *IEEE Access*, vol. 7, pp. 105402–105419, 2019.
- [48] O. Chughtai, N. Badruddin, A. Awang, and M. Rehan, "Congestion-aware and traffic load balancing scheme for routing in WSNs," *Telecommun. Syst.*, vol. 63, no. 4, pp. 481–504, 2016.
- [49] W. Ding, L. Tang, and S. Ji, "Optimizing routing based on congestion control for wireless sensor networks," *Wireless Netw.*, vol. 22, no. 3, pp. 915–925, 2016.
- [50] O. Banimelhem and S. Khasawneh, "GMCAR: Grid-based multipath with congestion avoidance routing protocol in wireless sensor networks," *Ad Hoc Netw.*, vol. 10, no. 7, pp. 1346–1361, 2012.
- [51] M. Tortelli, A. Rizzardi, S. Sicari, L. A. Grieco, G. Boggia, and A. Coen-Porisini, "S² DCC: Secure selective dropping congestion control in hybrid wireless multimedia sensor networks," *Wireless Netw.*, vol. 24, no. 1, pp. 309–328, 2018.
- [52] C. Sergiou, V. Vassiliou, and A. Paphitis, "Hierarchical tree alternative path (HTAP) algorithm for congestion control in wireless sensor networks," *Ad Hoc Netw.*, vol. 11, no. 1, pp. 257–272, 2013.
- [53] C. Sergiou, V. Vassiliou, and A. Paphitis, "Congestion control in wireless sensor networks through dynamic alternative path selection," *Comput. Netw.*, vol. 75, pp. 226–238, Dec. 2014.
- [54] W. Chen, Y. Niu, and Y. Zou, "Congestion control and energy-balanced scheme based on the hierarchy for WSNs," *IET Wireless Sensor Syst.*, vol. 7, no. 1, pp. 1–8, 2016.
- [55] K. Singh, K. Singh, L. H. Son, and A. Aziz, "Congestion control in wireless sensor networks by hybrid multi-objective optimization algorithm," *Comput. Netw.*, vol. 138, pp. 90–107, Jun. 2018.
- [56] V. Narawade and U. D. Kolekar, "ACSRO: Adaptive cuckoo search based rate adjustment for optimized congestion avoidance and control in wireless sensor networks," *Alexandria Eng. J.*, vol. 57, no. 1, pp. 131–145, 2018.

- [58] Y. Zhuang, L. Yu, H. Shen, W. Kolodzey, N. Iri, G. Caulfield, and S. He, "Data collection with accuracy-aware congestion control in sensor networks," *IEEE Trans. Mobile Comput.*, vol. 18, no. 5, pp. 1068–1082, May 2019.
- [59] J. Tan, W. Liu, T. Wang, S. Zhang, A. Liu, M. Xie, M. Ma, and M. Zhao, "An efficient information maximization based adaptive congestion control scheme in wireless sensor network," *IEEE Access*, vol. 7, pp. 64878–64896, 2019.



SUBHA KANTA SWAIN (Member, IEEE) received the Bachelor of Technology degree in electronics and communication engineering and the Master of Technology degree in ECE, with a focus on communication system engineering. He is currently pursuing the Ph.D. degree in wireless sensor networks with Siksha 'O' Anusandhan (Deemed to be University), Bhubaneswar, Odisha, India. His research interests include *ad-hoc* networks and wireless sensor networks. He has been

a Graduate Student Member of IEEE Communication Society, since 2014. He has been the Student Chapter Chair of SOA (Deemed to be University), since 2015.



PRADIPTA KUMAR NANDA (Senior Member, IEEE) received the bachelor's degree (Hons.) in electrical engineering from VSS University of Technology, in 1984, the master's degree in electronics systems and communication from NIT Rourkela, India, in 1989, and the Ph.D. degree in computer vision from IIT Bombay, in 1996.

From 1986 to 2007, he was a Faculty Member at NIT Rourkela, with the last position as a Professor and the Head of the Department, where

he has developed many laboratories. He was the Principal Investigator of the Center of Excellence on Industrial Electronics and Robotics. Thereafter he relocated himself to Siksha 'O' Anusandhan (Deemed to be University), Bhubaneswar, India, where he is currently serving as the Pro Vice-Chancellor. He is currently directing the research with the Image and Video Analysis Laboratory. He has visited U.K., USA, and Malaysia, in connection with academic programs. He was an Academic Visitor to the School of Computing, University of Leeds, U.K., in March 2006, and delivered lectures to their faculties and students. He has delivered several invited lectures in different international/national conferences and workshops/summer and winter schools. He has published 95 articles in various journals and conference proceedings, and has authored five research books and nine book chapters. He has supervised four doctoral students and 45 master's students. His research interests include image processing and analysis, bio-medical image analysis, video-tracking, computer vision, soft computing and their applications, and in ad-hoc wireless sensor networks.

Dr. Nanda is a fellow of IETE, India, a member IET, U.K., and a Life Member of ISTE, India. He has received the Bharat Jyoti Award from the Indian International Friendship Society, in 2012, and Odisha Technocrat Award on 15th September 2013 from Ever Green Forum of Odisha on the Engineers Day. He has been the program chair of three international conferences and the general chair of two international conferences. He has organized several national and international conferences. He has served as the Vice-Chair of IEEE Computational Intelligence Society, Kolkata Chapter, and the Chair of IEEE GRSS Society, Kolkata Chapter, India.